Response to Reviewers from Appendix G

Project No. S_Temp3462 (Continuation of S1051).

Project Title: Sustainable Practices, Economic Contributions, Consumer Behavior, and Labor

Management in the U.S. Environmental Horticulture Industry.

Administrative Advisor: Dr Craig Nessler

Proposed Termination Date: September 30, 2020.

Reviewer #1 (Rosson).

This is a well-developed research proposal that has potential to have positive impacts not only on the industry, but also to shape the policy debate related to the utilization and management of immigrant labor and its impacts on the US, regional, state and local economies. Strong methodology is being used that allows flexibility in analysis and clear development of results for presentation to appropriate stakeholders. It is encouraging to see such strong research/extension interface in the proposed project, allowing for the maximization of its impacts on appropriate audiences.

Recommendation: Approval.

Thank you very much for your review and comments.

Reviewer #2 (Ramirez).

A very detailed and ambitious project.

Recommendation: Approval.

Thank you very much for your review and comments.

Reviewer #3 (Safley)

This regional Review Committee has yielded good results in the past and should be continued.

Recommendation: Approval.

Thank you very much for your review and comments.